Author’s response to reviews

Title: CONNECT for Quality: Protocol of a Randomized Controlled Trial to Improve Fall Prevention in Nursing Homes

Authors:

Ruth A Anderson (ruth.anderson@duke.edu)
Kirsten Corazzini (coraz001@mc.duke.edu)
Kristie Porter (Kirstie.porter@duke.edu)
Katherine Daily (katherine.daily@duke.edu)
Reuben R McDaniel Jr. (Reuben.McDaniel@mccombs.utexas.edu)
Cathleen Colon-Emeric (Colon001@mc.duke.edu)

Version: 4 Date: 23 December 2011

Author’s response to reviews: see over
Respectfully,

[Signature]

Ruth A. Anderson, RN, PhD, FAAN
Virginia Stone Professor of Nursing
Research Development Coordinator, Office of Research Affairs
Senior Fellow, Center for the Study of Aging and Human Development
Duke University School of Nursing
Durham, NC 27710
Office: 919-668-4599
ruth.anderson@duke.edu